Author's response to reviews

Title: Cross-National Epidemiology of DSM-IV Major Depressive Episode

Authors:

Evelyn Bromet (ebromet@notes.cc.sunysb.edu)
Laura Helena Andrade (laurandrade@uol.com.br)
Irving Hwang (hwang@hcp.med.harvard.edu)
Nancy A Sampson (sampson@hcp.med.harvard.edu)
Jordi Alonso (jalonso@imim.es)
Giovanni de Girolamo (gdegirolamo@fatebenefratelli.it)
Ron de Graaf (rgraaf@trimbos.nl)
Koen Demyttenaere (koen.demyttenaere@uz.kuleuven.ac.be)
Chiyi Hu (chiyihu@yahoo.com.cn)
Noboro Iwata (iwatan@he.hirokoku-u.ac.jp)
Jagdish Kaur (jagdish.kaur@nic.in)
Aimee N Karam (aimeekaram@hotmail.com)
Stanislav Kostyuchenko (staskiev@yahoo.com)
Jean-Pierre Lepine (jean-pierre.lepine@lrb.aphp.fr)
Daphna Levinson (daphna.levinson@moh.health.gov.il)
Herbert Matschinger (Herbert.Matschinger@medizin.uni-leipzig.de)
Maria Elena Medina Mora (medinam@imp.edu.mx)
Mark Oakley Browne (mark.oakley-browne@dhhs.tas.gov.au)
Jose Posada-Villa (latos98@yahoo.com)
Maria Carmen Viana (mcviana@uol.com.br)
David R Williams (dwilliam@hsph.harvard.edu)
Ronald C Kessler (kessler@hcp.med.harvard.edu)

Version: 4 Date: 9 June 2011

Author's response to reviews:

This manuscript was previously revised and accepted by BMC Psychiatry, but was then transferred to BMC Medicine for publication based upon a recommendation from the editors. No changes have been made to this version.